Author’s response to reviews

Title: Wealth-related versus income-related inequalities in dental care use under universal public coverage: a panel data analysis of the Japanese Study of Aging and Retirement

Authors:

Keiko Murakami (mkeiko-tky@umin.ac.jp)

Hideki Hashimoto (hidehashimoto-circ@umin.ac.jp)

Version: 1 Date: 20 Oct 2015

Author’s response to reviews:

Dear Reviewers and Editorial Office

We are grateful for the opportunity to revise our manuscript “Wealth-related versus income-related inequalities in dental care use under universal public coverage: a panel data analysis of the Japanese Study of Aging and Retirement,” and the helpful comments of reviewers.

We have revised the manuscript according to the comments from editors and reviewers.

We hope that we have responded appropriately to the reviewers’ comments, and that our revised manuscript is now suitable for publication in BMC Public Health.